Author’s response to reviews

Title: Experimental infection of Balb/c nude mice with Hepatitis E virus

Authors:

Huang Fen (huangfen6789@163.com)
Zhang Wen (z0216wen@sjtu.edu.cn)
Gong Ga (gongga@sjtu.edu.cn)
Yuan Cong Li (ycl@sjtu.edu.cn)
Yan Yi Jia (y_yan@sjtu.edu.cn)
yang Shi Xing (yangshixing@sjtu.edu.cn)
Cui Li (lcui@sjtu.edu.cn)
Zhu Jian Guo (Zhu_jg@sjtu.edu.cn)
Yang Zhi Biao (zbyang@sjtu.edu.cn)
Hua Xiu Guo (hxg@sjtu.edu.cn)

Version: 6 Date: 13 May 2009

Author’s response to reviews:

Dear Editors,

Thank you for your letter. We are very pleased to learn that our manuscript is acceptable for publication. We appreciate you and all the reviewers’ help. In this revised manuscript, we carefully checked the manuscript formatting according to the template you supplied and added the corresponding author’s Email address in title page in line 11 of page 1, and keep the color of font consistent in the text. We do obtain the permission of Shanghai Research Center for Biomodel Organisms to perform this investigation. Thank you very much for your attention again. We are look forward to receiving your letter. Best Regards.

Sincerely yours,

Xiuguo Hua